Author's response to reviews

Title: A prospective study of dietary selenium intake and risk of type 2 diabetes

Authors:

Saverio Stranges (S.Stranges@warwick.ac.uk)
Sabina Sieri (Sabina.Sieri@istitutotumori.mi.it)
Marco Vinceti (marco.vinceti@unimore.it)
Sara Grioni (Sara.Grioni@istitutotumori.mi.it)
Eliseo Guallar (eguallar@jhsph.edu)
Martin Laclaustra (mlaclus@jhsph.edu)
Paola Muti (muti@ifo.it)
Franco Berrino (Franco.Berrino@istitutotumori.mi.it)
Vittorio Krogh (Vittorio.Krogh@istitutotumori.mi.it)

Version: 4 Date: 21 September 2010

Author's response to reviews: see over
September 21, 2010

Ms Robyn Hughes  
BMC Series Editorial Production Team

RE: MS # 2059743726396100 - A Prospective Study of Dietary Selenium Intake and Risk of Type 2 Diabetes

Dear Ms Hughes:

Please find enclosed our revised manuscript which has been edited as requested to conform to the journal’s editorial style.

Major revisions

A paragraph on Authors' contributions has been added after the Conclusions.

Minor revisions

The title has been modified using sentence case. We have placed the disclosures in the Acknowledgements section. We have inserted a heading titled 'Conclusions' after the discussion. We have removed all highlighting and tracking from the manuscript. The reference list has been modified as requested. We have removed the additional file from the manuscript. We have checked our manuscript for any typographical errors.

We would like to thank you for your assistance.

Please send all correspondence to my attention.

Look forward to hearing from you.

Sincerely,

Saverio Stranges, MD PhD  
Associate Professor of Cardiovascular Epidemiology  
Health Sciences Research Institute  
University of Warwick Medical School  
Medical School Building  
Gibbet Hill Campus  
Coventry CV4 7AL (UK)  
Tel: + 44 (0) 2476151153  
Fax: + 44 (0) 2476528375  
Email: S.Stranges@warwick.ac.uk